Author's Response To Reviewer Comments

Clo <u>s</u> e	
Dear Hans,	
Thank you! The tools have been registered. Thank you for spotting the typo in the data availability section. The materials have been submitted to the editorialmanager.	
Best wishes, Diogo	

Clo<u>s</u>e